Author's response to reviews

Title: Cerebral neurocysticercosis mimicking or comorbid with episodic migraine?

Authors:

Yannick F Fogang (yanfogang@yahoo.fr)
Massaman Camara (massamancamara@yahoo.fr)
Amadou G Diop (agalldiop@gmail.com)
Mansour M Ndiaye (nmansourndiaye@yahoo.fr)

Version: 2 Date: 8 March 2014

Author's response to reviews: see over
Type of article: CASE REPORT

Title of the article: Cerebral neurocysticercosis mimicking or comorbid with episodic migraine?

Covering letter

To,

The Editor of BMC Neurology

Subject: Submission of Manuscript for publication

Dear Sir,

We intend to publish an article entitled “Cerebral neurocysticercosis mimicking or comorbid with episodic migraine?” in your esteemed journal as a Case Report. In this case report, with clinico-radiological correlations, we discuss the possible causal influence of neurocysticercosis on the pathomechanisms of migraine. From a practical standpoint, this case report has potential clinical implications in the approach to patients presenting recurrent headaches in the setting of high neurocysticercosis prevalence.

On behalf of all the contributors I will act and guarantor and will correspond with the journal from this point onward.

Prior publication: nil

Support: nil

Conflicts of interest: nil

Permissions: nil
We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental thereto, exclusively to the journal, in the event that such work is published by the journal.

Thanking you,

Yours’ sincerely.